---
title: Next Steps
description: Where to go after building your first app
sidebar:
  order: 4
---

You've built your first Wails app and understand the basics. Here's where to go next.

## Learn More

Practice by building complete applications:

- [TODO App](/tutorials/02-todo-vanilla) - CRUD operations, state management, modern UI
- [All Tutorials](/tutorials/overview) - Notes apps, dashboards, system tray apps, and more

## Features

Wails provides native desktop capabilities:

- [Windows](/features/windows/basics) - Multiple windows, frameless windows, positioning
- [Menus](/features/menus/application) - Application menus, context menus, system tray
- [Dialogs](/features/dialogs/file) - File open/save, message dialogs
- [Events](/features/events/system) - Communication between components
- [Bindings](/features/bindings/methods) - Type-safe Go ↔ JavaScript calls
- [Clipboard](/features/clipboard) - Copy/paste operations
- [Drag & Drop](/features/drag-and-drop/files) - File drag and drop
- [Keyboard](/features/keyboard) - Global shortcuts

## Get Help

- [Discord](https://discord.gg/JDdSxwjhGf) - Ask questions, share projects
- [Examples](https://github.com/wailsapp/wails/tree/v3-alpha/v3/examples) - 50+ working examples
- [API Reference](/reference/overview) - Complete documentation
